Author’s response to reviews

Title: Combined antibacterial effect of essential oils from three most commonly used Ethiopian traditional medicinal plants on multidrug resistant bacteria

Authors:

Eshetu Dabale (gadisaeshetu@gmail.com)
Gebru Weldeargay (gebrumulu@gmail.com)
Kassu Tulu (kassudesta@gmail.com)
Getahun Dibaba (getahunthegeaye@gmail.com)
Sintayehu Ayena (sintayehu1981@gmail.com)
Kefiyeleuw Duffera (naftan2006@gmail.com)
Abera Dereressa (aberatekale91@gmail.com)

Version: 2 Date: 27 Nov 2018

Author’s response to reviews:

Author’s response to reviewers’ comments

Ref.: Ms. No: BCAM-D-18-01263R2

Title of manuscript: “Combined antibacterial effect of essential oils from three most commonly used Ethiopian traditional medicinal plants on multidrug resistant bacteria”

Authors’

1. Eshetu Gadisa, E-mail: gadisaeshetu@gmail.com
2. Gebru Weldearegay, E-mail: gabrumul@gmail.com
3. Kassu Desta, E-mail: kassudesta@gmail.com
4. Getahun Tsegaye, E-mail: getahunthegeaye@gmail.com
5. Sinteyehu Hailu, E-mail: sentayehu1981@gmail.com
6. Kefiyelewu Jote, E-mail: naftan2006@yahoo.com
7. Abera Tekle, E-mail: aberatakele91@gmail.com

Generally, in manuscript we used track change to incorporate or address the reviewers’ comments.

Response to Reviewer #2

1. “Are the methods appropriate and well described?”

Answers from Authors’: Yes. The methods and material part was appropriate and clear described in the research. However, there were grammatical errors through the manuscript.

2. Quality of written English: Not suitable for publication unless extensively edited

Answers from Authors’: Right, there are many grammatical and topological errors. As much as possible we tried to improve the errors using language experts and instructors that teaching at Addis Ababa University. Still it might have error. Thus, still we are open to take any concern and comments. We have accepted the comments and we tried to use available chance we have to edit it. Thank you again for your concern.